Author's response to reviews

Title: Continuous intraperitoneal insulin infusion in type 1 diabetes: a 6-year post-trial follow-up.

Authors:

Peter R. Van Dijk (p.r.van.dijk@isala.nl)
Susan J.J. Logtenberg (s.j.j.logtenberg@umcg.nl)
Klaas H. Groenier (k.h.groenier@umcg.nl)
Rijk O.B. Gans (r.o.b.gans@umcg.nl)
Nanne Kleefstra (n.kleefstra@isala.nl)
Henk J.G. Bilo (h.j.g.bilo@isala.nl)

Version: 2 Date: 31 March 2014

Author's response to reviews: see over
Reply to Editor

Title:
Continuous intraperitoneal insulin infusion in type 1 diabetes: a 6-year post-trial follow-up.

Authors:
Peter van Dijk, Susan Logtenberg, Klaas Groenier, Rijk Gans, Nanne Kleefstra and Henk Bilo

Manuscript ID:
1012917611242800

Corresponding author:
P.R. van Dijk
Diabetes Centre, Isala clinics
P.O. box 10400
8000 G.K. Zwolle, The Netherlands
Email: P.R.van.Dijk@isala.nl
Please state in the manuscript the full name of the ethics committee that approved the study.

First of all, we want to thank the editors for their valuable comments and consideration of the results of our study for publication in BMC Endocrine Disorders. In the revised manuscript we have added the full name of the local ethics committee:

“Approval by the medical ethics committee of the Isala (Zwolle, The Netherlands) was given for the crossover study in 2006 and the follow-up measurements in 2012.”